DIRECTIONS FOR ENTERING HOMEWORK SOLUTIONS WITH CAPA

CAP A is a “Computer-Assisted Personalized Approach for assignments, quizzes and examination”. It is strongly recommended that you answer the questions and work the problems before you ‘login’ to the computer. As you will see when you login, for some problems a :H (for Hint) shows up as an option on the screen once you have tried a problem.

Each assignment sheet has your Name and also a specific Personal Identification Number (PIN). Each set has a different (PIN). You may enter solutions either through vt100 emulation or through the web.

For TELNET:

1. Find a terminal and set the host to capa2.nscl.msu.edu. Different telnet packages have different startup procedures. In the basement of Kedzie, there are instructions for doing this. The CAPA login screen will appear.

2. At the login prompt enter the class name phy232s8.

login: phy232s8 3. Next, a new screen appears, enter your student number and the 4-digit PIN from your homework for your CAPA ID. Student Number: A********

CAPA ID: %#$@ You should now be logged in. Follow the options listed. For example enter T to try a problem set.

4. Be sure you log off correctly by typing :X to get credit for your set!

For WWW entry:

1. From your web browser, enter the address http://capa2.nscl.msu.edu/class.html 2. From the class list, pull down the menu and select phy232s8. Enter your student number and the 4-digit PIN from your homework for your CAPA ID.

3. Follow the instructions before you, and be sure to push the submit all answers button before you log off.

Entering answers For directions on entering answers, formats, units, etc., dial into http://theo15.nscl.msu.edu/physics232/handouts/capahelp.html